Author's response to reviews

Title: The electronic self report assessment and intervention for cancer: Promoting patient verbal reporting of symptom and quality of life issues in a randomized controlled trial

Authors:

Donna L Berry (donnalb@uw.edu)
Fangxin Hong (fxhong@jimmy.harvard.edu)
Barbara Halpenny (barbara_halpenny@dfci.harvard.edu)
Anne Partridge (ann_partridge@dfci.harvard.edu)
Erica Fox (erica_fox@dfci.harvard.edu)
Jesse R Fann (fann@uw.edu)
Seth Wolpin (swolpin@uw.edu)
William B Lober (lober@uw.edu)
Nigel Bush (nigel.e.bush.civ@mail.mil)
Upendra Parvathaneni (upendra@uw.edu)
Dagmar Amtmann (dagmara@uw.edu)
Rosemary Ford (rford@seattlecca.org)

Version: 3
Date: 8 May 2014

Author's response to reviews: see over
Authors' response to reviews

Title: The electronic self report assessment and intervention for cancer: Promoting patient verbal reporting of symptom and quality of life issues in a randomized controlled trial

Authors:
- Donna L Berry: donnalb@uw.edu
- Fangxin Hong: fxhong@jimmy.harvard.edu
- Barbara Halpenny: barbara_halpenny@dfci.harvard.edu
- Anne Partridge: Ann_Partridge@dfci.harvard.edu
- Erica Fox: erica_fox@dfci.harvard.edu
- Jesse R Fann: fann@uw.edu
- Seth Wolpin: swolpin@uw.edu
- William B Lober: lober@uw.edu
- Nigel Bush: nigel.e.bush.civ@mail.mil
- Upendra Parvathaneni: upendra@uw.edu
- Dagmar Amtmann: dagmara@uw.edu
- Rosemary Ford: rford@seattlecca.org

Version: 3 Date: 8 May 2014

Reviewer: Roxanne Jensen

Reviewer's report:

1) Please include some basic information about the SDS-15 scale in the methods section.
   This information has been added to page 6.

2) Small formatting fix -- Page 6. The first and second reference to the SDS-15 should be fixed.
   Corrected

3) Table 1: SDS-15 scores seem to be repeated across both groups.
   Corrected

4) In the methods section the multivariate model is said to include: age, clinical service, working status and baseline SDS-15. Table 3 lists the multivariate model with only age, service, work status. Please revise table three to include baseline SDS-15 scores.
   Added to Table 3 as requested and edited.